Author's response to reviews

**Title:** The relationship between low maternal serum vitamin D levels and glycemic control in gestational diabetes mellitus assessed by HbA1c levels

**Authors:**

- Ahmed El lithy (ahmedellithy75@gmail.com)
- Rana M.A. Abdella (rana.abdella@yahoo.com)
- Yahia M El-Faissal (yahiaelfaissal@gmail.com)
- Ahmed M Sayed (ahmad.mahmoud.ali@hotmail.com)
- Rasha M Abdel Samie (drrasha76@gmail.com)

**Version:** 4  
**Date:** 1 October 2014

**Author's response to reviews:** see over
Cairo, Egypt  
Wednesday, October 1st, 2014  
From: Dr Yahia El-Faissal, MD  
To: Dr Peter O'Donovan,  
Executive Editor, BioMed Central  
Pregnancy & Childbirth Journal  

Dear Dr O'Donovan;  

On behalf of all my colleagues, I would like to express my gratitude to the editorial board for their sincere work and valuable comments; we are delighted that you found our study suitable for publication in your esteemed journal. As concerns the editorial comment stated in your last letter, I am pleased to give you our reply on it as follows:  
The editorial comment was: “Please indicate full name of ethics review board that approved the study and written informed consent used in the study in the methods section.”  
First: the full name of the board was added to the methods section (lines 111-113).  
Second: details of the written informed consent used in this study were added (lines 130-134).  
We hope that these modifications are acceptable, looking forwards to receive your reply soon. Thank you again for your efforts and please accept our best regards.  

Yours sincerely;  

Dr Yahia El-Faissal, M.D.  
Lecturer, OBGYN department  
Cairo University